Author's response to reviews

Title: A prospective, randomized, double-blinded single-site control study comparing blood loss prevention of tranexamic acid (TXA) to epsilon aminocaproic acid (EACA) for corrective spinal surgery

Authors:

Kushagra Verma MS (vermak01@med.nyu.edu)
Thomas J Errico MD (Thomas.Errico@nyumc.org)
Kenneth M Vaz BS (kenneth.vaz@gmail.com)
Baron S Lonner MD (BLonner@nyc.rr.com)

Version: 4 Date: 25 February 2010

Author's response to reviews:

February 24th 2010

Dear Reviewers,

You comments for the following submitted manuscript (citation below) were appreciated and well received. We have attached a revised version of the manuscript with a point by point response to the reviewers concerns as well.

MS: 1180533212908659

A prospective, randomized, double-blinded single-site control study comparing blood loss prevention of tranexamic acid (TXA) to epsilon aminocaproic acid (EACA) for corrective spinal surgery Kushagra Verma, Thomas J Errico, Kenneth M Vaz and Baron S Lonner

Reviewer 1:

Competing interests - Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'.

The authors have all stated that they have no competing interests in the publication of this manuscript. We have included this directly after the conclusions section and before Authors' contributions.